Author's response to reviews

Title: Correlation between human papillomavirus infection and bladder transitional cell carcinoma

Authors:

Mohammadreza Barghi (h_sasan@hotmail.com)
Amir Hajimohammadmehdiarbab (h_sasan@hotmail.com)
Seyed Mohammad Mehdi Hosseini Moghaddam (h_sasan@hotmail.com)
Bahram Kazemi (h_sasan@hotmail.com)

Version: 4 Date: 18 October 2005

Author’s response to reviews:

Dear Dr. Holmes:

We revised our manuscript as you suggested and title, references, manuscript sections, and tables were revised as needed. Two new sections, competing interests and authors’ contributions, were added. Please do not hesitate to contact us if further information is necessary.